Author’s response to reviews

Title: Distinctive spinal changes in two patients with unusual forms of autosomal dominant endosteal hyperostosis: two case reports

Authors:

Ali Al Kaissi (ali.alkaissi@osteologie.at)
Franz Varga (franz.varga@osteologie.at)
Elisabeth Zwettler (elisabeth.zwettler@osteologie.at)
Shahin Zandieh (shahin.zandieh@osteologie.at)
Klaus Klaushofer (klaus.klaushofer@osteologie.at)
Franz Grill (franz.grill@oss.at)

Version: 5 Date: 6 August 2007

Author’s response to reviews: see over
Dear Editor,

Re-our case report entitled “Distinctive spinal changes in two patients with unusual forms of autosomal dominant endosteal hyperostosis: two case reports”

We followed the comments of the Reviewers point by point as clarified in our attached document (response to authors)
As you know, the authors are not Native English speakers, however, we did our best to correct some errors.
We hope that our modifications and additions are sufficient to get our paper published.

Sincerely yours
Al Kaissi